\magnification 1200
\def\\{\vskip10pt}\footline{}
\parindent=0pt
\long\def\moose#1@#2@@{\hbox{\hbox to .5\hsize{\vtop{#1}\hss}\vtop{#2}}}

\centerline{\bf Math 105{--}Fall 1991}
\centerline{\bf Syllabus}
\bigskip

Course instructors are Anderson, Hu, Smyth and Taylor.

The secretary for the course is Connie E.\ Slack.  


Below is a suggested
schedule of sections to cover.
It is important to prepare the students for
the exams.
Each of us is responsible for preparing one of the multiple
choice exams.
When it is your turn to prepare the exam please have a copy to Taylor
by the indicated date.
Preliminary copies will be distributed to all of us for comments before
the final version is prepared.
\bigskip

\centerline{{\bf Schedule}}

\medskip
\newdimen\aa\newdimen\bb
\aa=30pt\bb=50pt
\long\def\a#1 #2 {\hbox{\hbox to\aa{#1\hss}\hbox to\bb{#2\hss}}}
\long\def\b#1 #2 #3.{\hbox{\hbox to\aa{#1\hss}\hbox to\bb{#2\hss}\hbox{#3.}}}


\moose

\null
\a8/28 0.1 
\a8/30 0.2--0.3 
\\
\b9/2 0.4--0.5 9/3 Test I Smyth. 
\a9/4		1.1
\a9/6		1.2
\\
\a9/9	1.3
\a9/11	1.4
\a9/13	1.6
\\
\b9/16	Review			9/17 Exam I.
\a9/18	Discuss			
\a9/20 1.7
\\
\a9/23	1.8
\a9/25	2.1
\a9/27	2.2 
\\
\a9/30	2.3
\b10/2	2.4				10/2 Test II Hu. 
\a10/4	2.5
\\
\a10/7	2.6
\a10/9	2.6
\a10/11	3.1
\\
\a10/14	3.2
\b10/16	Review			10/17	Exam II \& Midterm Grades.
\a10/18	Discuss
\\
\b10/21--10/25 {}  Fall  Break.
\\ @
\a10/28	3.3
\a10/30	3.3
\a11/1		4.1
\\
\a11/4	4.2
\a11/6	4.3
\a11/8	4.4
\\
\b11/11	4.5				11/12  Test III Anderson. 
\a11/13	4.6
\a11/15	5.1
\\
\a11/18	5.1
\a11/20	5.2
\a11/22 5.2
\\
\b11/25	Review		Exam III 11/26.
\a11/27	Discuss
\a11/29 Thanksgiving
\\
\a12/2	6.1 
\b12/4	6.2					12/4	Final Taylor.
\a12/6	6.3
\\
\a12/9	6.3
\a12/11	Review
\a12/13	Review
\\
\a12/16 Final 
@@


\end


